
GSN Data Quality 
NGS Date Date Overall User Other notes

Net ID upgrade Site DAS Comm Z N/1 E/2 Z N/1 E/2 Z N/1 E/2 yr jd Problem ID
Expected 
Solution Quality

Feed 
back

II AAK 1 2010 180 1

II ABPO 1 2010 279 1 STS1-V replaced; STS1-E very noisy

IU ADK 1 9/30/09 1 Upgraded 07/09.  Power problems compromise data availability.  Several different solutions under consideration.

IU AFI 1
10-
Z,1 1 Upgraded 03/10.  Problems with triggering algorithm.  Vault is leaking and requires repair.

IU ANMO 1 1 Upgraded 06/08. Problems with triggering algorithm

IU ANTO 1 00-1,2 1 Upgraded 07/10 and a 10 sensor was added.  Primary sensor is still a KS-36000

CU ANWB 1 1

II ASCN 1 2010 138 2/28/10 open 1 Episensor has gain issue

CU BBGH 1 1

IU BBSR 1  1 Upgraded 08/08.  Initial grounding problems with Episensor appear to be resolved.

CU BCIP 1 1

IU CHTO 1 10 1 Upgraded 09/09. 

II CMLA 1 2010 295 10/22/10 Q4/2011 1 Need to replace failed CMG-3T.

II COCO 1 2010 302 1

IU COLA 1 1 Upgraded 07/09. E/W channel of the accelerometer has high-noise but is working.

IU COR 1 1 Upgraded 09/09.  Problems with triggering algorithm.

II EFI 1 2010 209 1 FBA now OK

II ESK 1 2010 58 1

II FFC 1 2011 20 1 STS1-E is noisy

IU FUNA 1      1 Upgraded 1/11.  Response issue with the H1 component of the Episensor.

IU GNI 1    1 Upgraded 10/10, which corrected the overdamped corner issues on the STS-1's

IU GRFO 1 1 Q330 was replaced on 03/02/2010 which solved the gapping problem. Primary sensor is a KS-36000.

CU GRGR 1 1

CU GRTK 1 1

CU GTBY 1 1

IU GUMO 1 10 1 Upgraded 01/10. Problems with triggering algorithm

IU HNR 1 10 04/2010 1
Upgraded 04/10. The 54000 is stuck in the borehole, so currently there are two STS-2's on site.  The "primary" 
STS-2 is noisy but will be replaced with KS-54000 when we can return.

IU HRV 1 1 Upgraded 09/08.  Recurrence of overdamping problem resolved 05/10.

IU JOHN 1 1 Upgraded 08/10 with a new power system and a Trillium T240 as a secondary sensor.  

II JTS 1 2010 230 3/7/10 Q2/2011 1 New pier to be built Q1 2011; noise pulses on STS1-V.

II KAPI 1 2009 321 1

IU KBL 1 1 We are not directly involved in the operation of this station.  It is performing well.

IU KBS 1 1
Upgraded 06/10. 1db difference in the lp spectra between 10 and 00  horizontals for Japan quake.  Cal data 
looks good so we need to investigate further.

IU KEV 1 10 1 Upgraded 09/08.  Problem with triggering algorithm.

IU KIP 1 8/25/10 1
Upgraded 08/10 which corrected most of the STS-1 overdamping issues; Z component is still undergoing 
evaluation.  The comm link is prone to backing up.  We are planning on installing cell modems

IU KMBO 1   1 Upgraded 04/10 which corrected the overdamped corner issues.

IU KNTN 1   12/4/07 1
This station has been plagued by power problems since it was installed.  Althought a secondary sensor is 
installed, it is used as a backup sensor.  ASL trained volunteers will visit site 5/11.

IU KONO 1 10 9/12/09 1 Upgraded 09/09. Rail line in the mine and is a noise source.  There is some pusling on the STS-1 H1.

IU LSZ 1 1 Upgraded 09/08

IU LVC 1 1 Upgraded 04/09. Com dropped out 03/11 due to a problem with the CTBTO VSAT system.

IU MACI 1 1 Problems with gappy data appear to be improving. Host controls station.

IU MAJO 1 00,10 1 Upgraded 11/09. STS-1 ?H1 is a little noisy; we need to replace the baseplate during the next site visit.

IU MAKZ 1 1 Station partially upgraded 09/10.  STS-1s were not restored and will be addressed at a later date.

CU MTDJ 1 1

II NNA 1 2011 68 1

IU PAB 1  10 1 Upgraded 08/09.  Site visit in 2010 to replace cables resolved the remaining problems with overdamping.

II PALK 1 2009 342 7/30/10 unknown 1 No date set for CMG replacement

II PFO 1 2010 255 1

IU PMG 1 10 1 Upgraded 04/10.  Local power is not highly reliable.  Seismometer vault is leaking.

IU POHA 1 1 Upgraded 1/11.

IUQSPA (00,10) 1 1 Upgraded 1/11.

IU RAR 1 3/9/10 1
Upgraded 03/10. KS-54000 replaced in Jan 2011.  We see a difference in the lp response of the horizontals 
between the two instruments.  Need to do further analysis to determine the issue.

IU RCBR 1     1 Upgraded 08/09. 

IU RSSD 1 10-Z 1 Upgraded 09/09.

CU SDDR 1 1

IU SDV 1     1 Upgraded 01/10.   Problem with overdamped corner on STS-1 ?H1.  

IU SFJD 1 10-Z 1 Upgraded 09/08.

IU SJG 1 12/11/09 1 Upgraded 04/10.

IU SLBS 1 1 Station has frequent outages due to low solar conditions

IU SNZO 1 1 Upgraded 12/10.  The problematic CMG-3 was replaced with and STS-2

IU SSPA 1        1 Upgraded 10/10.

II SUR 1 2010 178 1

IU TARA 1 10 1 Upgraded 11/09.  STS-2 replaced 1/11.

IU TATO 1 00 1 Upgraded 09/09.

II TAU 1 2011 20 1

CU TGUH 1 3 6 1

IU TRQA 1     10-1 1 Upgraded 05/09.  Some sporadic problems with communications.

IU TSUM 1 1 Upgraded 09/08.

IU TUC 1 10-Z 1 Upgraded 05/09.

II UOSS 1 2011 19  Q1/2011 1 Pinging STS2 replaced with new STS2.5 1/9/2011

IU WAKE 1 1
Upgraded 11/07.  Site is prone to corrosion.  Station is currently experiencing a timing problem and a new GPS 
unit and antenna has been shipped.

IU WCI 1 3/25/10 1 Upgraded 09/10.

II WRAB 1 2009 302 unknown 1 Solar array needs replacement.

IU WVT 1 1
Upgraded 01/11. Temporary STS-1 Z was installed for 50 days while we worked on the SLU instrument.  The 
original STS-1 Z is currently installed.

II ALE 2003 237 1 Data not yet received
II ARU 2009 315 1 FBA again OK

II BFO 2011 1 1 STS1 clipped due to very high gain of BFO's custom pre-amplifier

IU BILL 1
Site has chronic noise from permafrost.  We have suggested relocating the site.  There are problems with a 
conflicting IP address that introduce gaps in the data.  No 40sps data telemetered in real time.  Unable to run 

II BORG 2005 25 1

IU CASY 3/25/10 1
STS-1 horizontals have severly lowered lp responses. N/S FBA is dead.  Vault is prone to flooding.  New STS-2 
was installed 02/11.  Hope to upgrade winter 2011-2012

IU CTAO 1

II DGAR 2008 57 1 Episensor has gain issue

II HOPE 2005 287 1

II KDAK 2009 223 1

IU KIEV 1 STS-1 horiztonals are pulsing.

II KIV 1998 194 1

IC KMI 4/2008 1
No access to real-time data from this station. After a large event on 2008,133,06:06 UTC, the E/W component of 
the STS-2 is showing a large decrease in sensitivity. It looks to be around 3-4 dB.  Working with hosts to arrange 

II KWAJ 2003 8 1 STS2 clipped, both high and low gain

IU LCO 1

II MBAR 2007 300 3/10/07 2012 1 KS-54000 Z is inoperable; CMG-3T high gain clipped, low gain on scale

IU MBWA 10 1

IC MDJ 00,10 1
Slight sensitivity differences between STS-1s and STS-2. Station needs SensOrLoc analysis. Working with hosts 
to arrange calibration.

IU MIDW 00 1 No low-gain sensor present. 

IU NWAO 11/6/09 1 KS-54000 was replaced and station is back up as of 10/05/2010

II OBN 1998 169 1

IU PET 1

IU PMSA 1 No low-gain sensor present.

IU PTCN 1 Occasional problems with communications.

IU SBA 1 No low-gain sensor present.

II TLY 1998 299 1

IU TRIS 1 Some spiking on FBA-23

IC WMQ 8/28/09 1 No access to real-time data from this station. Working with hosts to arrange calibration.

IU YAK 1 40sps data not telemetered in real time.

IU DWPF 1 Apr-11 2 Upgraded 09/10. KS-54000 is inoperative and the T240 is pulsing.

IU INCN 1 10 8/16/10 2
Upgraded 11/09. Vault floods periodically.  Most recent incursion of water in Oct 10 has damaged the STS-1s.    
Working with the station operator to install a sump pump and fix STS-1 problems.

IU MA2 1    10 2
Station upgraded by hosts 07/10.  STS-1 is producing noise; STS-2 appears to be ok.  Have not seen any 
Episensor data.  Vault looks wet in station photographs.  No 40sps data telemetered in real time.

IU OTAV 1    5/25/10 2 Upgraded 04/09.  The STS-1 ?H1 is pulsing and the vertical still shows a slightly lowered response

IU PAYG 1    2010,201 2 Upgraded 05/10.  Large LP pulsing problem with the ?H2 channel on the KS-54000 started in July.

IU PTGA 1    2
Upgraded 11/09.  KS-54000 noisy - and stuck in hole.  Replacement sensor is on-site and waiting for a 
maintenance trip.

II RAYN 1 2010 44 2/1/10 unknown 2 Borehole is leaking at 90m depth.

IU SAML 1     2 Upgraded 11/09. Sensitivity issues on the Z and 1 Need to replace KS-54000.

II ABKT 2009 339 Q2/2011 2 Data not yet received
IC BJT

00-
Z,1 8/1/07 2  Lowered lp response for STS-1's.  Unable to run calibrations.

II BRVK 2007 262 Q4/2010 2 Instruments to be relocated to new vault by 2012.

IU DAV 2
Lowered lp response for STS-1s.  We plan to train the station operators in FY11 and have them upgrade the 
station.  There have been some problems with the vault that we are attempting to repair

IC ENH 10 11/20/09 2
FBA is improved, but the N/S component is dead.  Major problem with STS-1 gains.  STS-2 installed in March 
2011 (after Japan event).

II ERM 2006 242 2/28/10 open 2 Serious noise issues with STS1, all components; STS1s clipped for this event

IU FURI 2 Scheduled for an upgrade in FY11.

IC HIA 10/18/10 2
FBA is present but looks bad. We suspect an overdamped corner on the STS-1Z.  Replacement parts have been 
received.  Working with hosts to arrange calibration.

IU HKT 4/19/10 2
The salt-mine site is an extremely hostile location for all equipment.  Pier shifts constantly.  Upgrade planned in 
FY11.

II KURK 2005 153 Q4/2010 2

IC LSA 10 2
Pulsing on the STS-1 horizontals; STS-2 E/W down since 04/2006; FBA N/S down since 08/2009; sensitivitiy 
issues between STS-1 vs. STS-2 N/S.    Working with hosts to arrange calibration.

II LVZ 2000 165 2 Data system requires frequent power resets resulting in highly gapped data

IC QIZ 0 3/24/10 2 Working with hosts to arrange calibration.

IU RAO 8/5/09 2 STS-1Z is dead.  STS-1 is vulnerable to temp changes and should be replaced with different instrumentation.

IC SSE 2
No access to real-time data from this station. Gain issue for either primary or secondary instruments; needs 
SensOrLoc analysis. Working with hosts to arrange calibration.

IU TEIG 10/12/06 2 All senors have gain issues.  Working with UNAM & CTBTO to develop an upgrade plan.

IU TIXI 2
Site has chronic noise from permafrost.  Currently having some problems with data acquisition unit.  40sps data 
not telemetered in real time.

IU ULN ? ? 2 LOWERED LP RESPONSE ISSUES ON STS-1 HORIZONTALS

IC XAN 9/11/08 2 No access to real-time data from this station. FBA inop. Working with hosts to arrange calibration.

IU XMAS 10 7/21/10 2 The primary STS-2 is down as of 07/21/2010

IU YSS 0 2 STS-1 Z has 360 s ringing.  40sps data not telemetered in real time.

IU CCM 12/27/08
2nd Q 

2011 3 Station is down.  Sensor status is unknown.  Cannot be upgraded until late spring 2011.

IU KOWA 3 Vault environment is highly corrosive.  Station to be re-installed FY11, actual schedule depends on CTBTO.

II MSEY 2007 128 1/14/11 4/3/54 3 Mains power severed; local company has promised tor restore lines

IU MSKU 3
Station vulnerable to vandalism as well as power problems.  Working with CTBTO on possible solutions.  
Expected to be upgraded in FY11

II MSVF 2010 203 1/2/11 2/24/11 3 Calibration successful for vertical only.  GPS clock failure; replacement sent.

II NIL 1997 141 5/3/09 Q2/2011 3 Station upgrade awaiting restoration of telemetry by AFTAC

II NRIL 1992 351 2/1/97 unknown 3 Station repair awaiting upgrade by EME.

II RPN 2002 331 12/17/10 Q1/2011 3 Recording building needs replacement; VSAT repaired; GPS clock failure

II SACV 2008 152 8/23/10 3/15/11 3 DAS failure

II SHEL 1995 177 1/14/11 4/3/11 3 Vault power supply has failed; replacement expected to arrive in St. Helena Apr 3rd

Legend
DATA AVAILABLE AND NO ISSUES FOUND

DATA AVAILABLE WITH ISSUES

NO TELEMETRY FROM THIS STATION

NO SENSOR INSTALLED

NO DATA/BAD DATA

UPGRADED TO NG SYSTEM

Overall Data Quality Statistics

quality # % # %
1 103 73% 74 90%
2 29 20% 8 10%
3 10 7% 0 0%

total 142 82
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